@AGUPUBLICATIONS

Geophysical Research Letters

RESEARCH LETTER

10.1002/2017GL074662

Key Points:

- We demonstrate that the volcanic aerosol effect is the primary signal dominating the pattern of decadal variability in the BD circulation
- Main difference between observations and climate models concerns changes in the depth and strength of the aerosol effect on BD circulation
- We demonstrate that Northern Hemisphere slowdown is partly driven by minor volcanic eruptions after 2008

Supporting Information:

Supporting Information S1

Correspondence to:

M. Diallo, m.diallo@fz-juelich.de

Citation:

Diallo, M., Ploeger, F., Konopka, P., Birner, T., Müller, R., Riese, M., ... Jegou, F. (2017). Significant contributions of volcanic aerosols to decadal changes in the stratospheric circulation. *Geophysical Research Letters*, *44*, 10,780–10,791. https://doi.org/10.1002/2017GL074662

Received 20 JUN 2017 Accepted 11 OCT 2017 Accepted article online 16 OCT 2017 Published online 30 OCT 2017

©2017. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Significant Contributions of Volcanic Aerosols to Decadal Changes in the Stratospheric Circulation

M. Diallo^{1,2}, F. Ploeger¹, P. Konopka³, T. Birner³, R. Müller¹, M. Riese¹, H. Garny⁴, B. Legras², E. Ray⁵, G. Berthet⁶, and F. Jegou⁶

¹Institute of Energy and Climate Research, Stratosphere (IEK-7), Forschungszentrum Jülich, Jülich, Germany, ²Laboratoire de Météorologie Dynamique, UMR8539, IPSL, UPMC/ENS/CNRS/Ecole Polytechnique, Paris, France, ³Department of Atmospheric Science, Colorado State University, Fort Collins, CO, USA, ⁴Deutsches Zentrum für Luft-und Raumfahrt, Institut für Physik der Atmosphäre, Oberpfaffenhofen, Germany, ⁵Chemical Sciences Division, Earth Systems Research Laboratory, NOAA, Boulder, CO, USA, ⁶Laboratoire de Physique et Chimie de IEnvironnement et de l'Espace (LPC2E), UMR 7328 CNRS-Université dOrléans, Orléans, France

Abstract The stratospheric circulation is an important element of climate as it determines the concentration of radiatively active species like water vapor and aerosol above the tropopause. Climate models predict that increasing greenhouse gas levels speed up the stratospheric circulation. However, these results have been challenged by observational estimates of the circulation strength, constituting an uncertainty in current climate simulations. Here, we quantify the effect of volcanic aerosol on the stratospheric circulation focusing on the Mount Pinatubo eruption and discussing further the minor extratropical volcanic eruptions after 2008. We show that the observed pattern of decadal circulation change over the past decades is substantially driven by volcanic aerosol injections. Thus, climate model simulations need to realistically take into account the effect of volcanic eruptions, including the minor eruptions after 2008, for a reliable reproduction of observed stratospheric circulation changes.

Plain Language Summary The upper atmospheric circulation is an important element in the climate system as it determines the distributions and lifetimes of key greenhouse gases and impacts the Earth's radiation budget and surface climate. Current climate models rather uniformly predict that increasing greenhouse gas levels speed up the upper atmospheric circulation. However, these results contrast with observations, constituting a major uncertainty in current climate simulations. Our paper shows that the observed pattern of decadal circulation change over the past few decades is substantially driven by volcanic aerosol injections. The recently observed slowdown of the stratospheric circulation in the Northern Hemisphere is by 50% attributable to stratospheric aerosol from minor volcanic eruptions after 2008, which should no longer be neglected in climate simulations.

1. Introduction

The stratospheric circulation, known as the Brewer-Dobson circulation (Bönisch et al., 2011; Butchart, 2014), is an important element of climate as it determines the lifetime and concentration of key greenhouse gases, such as H_2O , CH_4 , and O_3 , as well as the amount of stratospheric water vapor (Riese et al., 2012; Solomon et al., 2010) and aerosol above the tropopause (Solomon et al., 2011). Climate models predict that increasing greenhouse gas levels speed-up the stratospheric circulation (e.g., Butchart et al., 2010; Garny et al., 2011).

However, these climate model results have been challenged by observational estimates of the circulation strength (Engel et al., 2009; Haenel et al., 2015; Mahieu et al., 2014; Ray et al., 2014), constituting an uncertainty in current climate simulations. Recent model simulations (Diallo et al., 2012; Ploeger, Riese, et al., 2015) driven by meteorological reanalysis (e.g., Dee et al., 2011; Kobayashi et al., 2015) suggest that the pattern of stratospheric circulation change is more complicated than a uniform speedup. The stratospheric circulation changes over the last decades derived from in situ and satellite measurements of trace gases (Engel et al., 2009; Stiller et al., 2012) are consistent with these model simulations. This consistency indicates an important

role of natural variability (e.g., quasi-biennial oscillation (QBO), El Niño – Southern Oscillation (ENSO), and volcanic eruptions) for decadal circulation change. The natural variability is strongly constrained by assimilation in the meteorological reanalysis data but difficult to represent reliably in climate models.

Volcanic aerosols from major eruptions (e.g., Pinatubo in June 1991) are well known to have a significant effect on global climate, particularly on stratospheric ozone and temperature (Hansen et al., 1996; Tilmes et al., 2011). The stratospheric aerosol originates from injected carbonyl sulfide (OCS) and sulfur dioxide (SO₂) oxidized to sulfuric acid (H_2SO_4), which then condensates to form a layer of fine sulfate aerosol droplets (H_2SO_4/H_2O) also known as the *Junge Layer* (Junge & Manson, 1961). In particular, during the major eruption of Mount Pinatubo, large amounts of aerosol were directly injected into the tropical lower and mid stratosphere (e.g., Vernier et al., 2011). Stratospheric aerosol particles induce cooling at the surface by reflecting some of the incoming solar energy back to space and heating in the stratosphere by absorbing some of the solar radiation and near-infrared radiation emitted by the Earth's surface. Recently, the volcanic aerosol injection into the extratropical lowermost stratosphere from the post-2008 minor volcanic eruptions (in contrast to Pinatubo) was found to significantly contribute to the stratospheric volcanic aerosol optical depth (AOD) (Andersson et al., 2015; Fromm et al., 2014; Vernier et al., 2011). Even the aerosol contribution from a series of minor volcanic eruptions during the first decade of the twenty-first century has been shown to have the potential to cause significant surface cooling (Solomon et al., 2011) and likely contributed to the recent decadal fluctuations in global warming (Huber & Knutti, 2014; Santer et al., 2014; Schmidt et al., 2014).

Here we attempt to quantify the impact of volcanic eruptions on changes in the pattern of the stratospheric circulation over the past few decades. We particularly focus on the Pinatubo eruption, which shows the clearest and most persistent signal in the lower stratosphere (see Table S1 in the supporting information). We describe the observational data record, the mean age, and multilinear regressions in section 2. Section 3 contains evidence for the impact of volcanic eruptions on stratospheric mean age and its trends. Finally, we discuss our results in the context of the puzzling discrepancy between climate models and observations regarding a potential acceleration of the stratospheric BD circulation.

2. Method and Data

2.1. Models

The results presented here are based on calculated mean age of stratospheric air (see section 2.2 for a definition) and observational stratospheric AOD from satellites. We used mean age calculated with CLaMS (Ploeger, Riese, et al., 2015) and TRACZILLA (Diallo et al., 2012) models, both driven by meteorological reanalysis data. Both simulations used 3-hourly horizontal winds and diabatic heating rates, as the vertical velocity in isentropic coordinates, both from ERA-Interim (ERA-I) (Dee et al., 2011) and Japanese 55 year Reanalysis (JRA-55) (Kobayashi et al., 2015) provided by the European Centre for Medium-Range Weather Forecasts and the Japan Meteorological Agency.

2.2. Mean Age of Air

Mean age in the CLaMS model is calculated from a "clock tracer" that is an inert tracer with a linear increase in the lowest model layer at the surface (Hall & Plumb, 1994). Further details about the CLaMS model and the specific simulation are described by Ploeger, Riese, et al. (2015) and Pommrich et al. (2014). For comparison, we consider mean age calculations from the Lagrangian trajectory model, TRACZILLA (Legras et al., 2005), using the setup described by Diallo et al. (2012, 2017) . Both CLaMS and TRACZILLA mean age simulations have been thoroughly validated with balloon-borne measurements and Michelson Interferometer for Passive Atmospheric Sounding (MIPAS) satellite observations of CO_2 and SF_6 , giving confidence in their robustness (Diallo et al., 2012; Ploeger, Riese, et al., 2015). Even regarding decadal variations in the mean age, there is agreement between the simulations and observations, both for the hemispheric asymmetry in the pattern of the circulation change during the last decade (2002–2011) and for the long-term weak increase of mean age in the Northern Hemisphere upper stratosphere (Diallo et al., 2012; Ploeger, Riese, et al., 2015).

2.3. Stratospheric Aerosol Optical Depth

Global stratospheric AOD is estimated from the monthly mean extinction ratio profile, which is the ratio between the particulate and molecular extinction, and is a proxy for the aerosol mixing ratio. Here we use AOD based on SAGE II (Stratospheric Aerosol and Gas Experiment) (1985–2005), on GOMOS (Global Ozone Monitoring by Occultation of Stars) (September 2005 to May 2006), and on CALIPSO (CloudSat and the Cloud-Aerosol Lidar and Infrared Pathfinder Satellite Observation) (June 2006 to December 2012)



Figure 1. Globally averaged time series of the stratospheric AOD, deseasonalized mean age, and residual of the multiple linear regression with and without removal of all AOD signal. (a) Stratospheric AOD time series is averaged from 50°S to 50°N over the 1989–2012 time period and is shown for merged satellites data sets (GISS: black and SAGE II+GOMOS(525 nm)+CALIPSO (532 nm): red, blue, and green). (b) The deseasonalized mean age driven by ERA-I and JRA-55 reanalyses is globally averaged between 72°S and 72°N and 16–28 km. (c and d) The residual of the multiple linear regression with (red dashed line) and without (black dashed line) removing the AOD signal from the deseasonalized mean age (Figure 1b). The gray shading area indicates the standard deviation.

measurements (red line in Figure 1a). In order to gain confidence in the robustness of our results, we have also used other global stratospheric AOD data sets from the Goddard Institute for Space Studies (GISS) (Sato et al., 1993) and SAGE II merged with OSIRIS (Optical Spectrograph and Infrared Imaging System) (Bourassa et al., 2012).

Figure 1a shows the global stratospheric AOD from different merged satellite observations (Sato et al., 1993; Vernier et al., 2011). Most noticeable is the strong peak in aerosol loading between June 1991 and December 1997 associated with the Pinatubo eruption. The Pinatubo plume instantaneously reached the stratosphere because of its high intensity and tropical location and was further uplifted by enhanced tropical upwelling during the easterly phase of the quasi-biennial oscillation (QBO) (Trepte & Hitchman, 1992). The long lasting aerosol enhancement related to the Pinatubo eruption masked the impact of other weaker eruptions with stratospheric influence occurring during this period, including Cerro Hudson in October 1991 and Rabaul in Papua New Guinea in September 1994. The 1998–2002 period shows no discernible volcanic enhancement in the stratosphere and will be referred to as "quiescent period" in the following. There is only a small and slow rise in stratospheric AOD from 2002 to 2005 followed by two tropical volcanic eruptions (Soufrière Hills on 20 May 2006 and Tavurvur on 7 October 2006) that lead to the peak in AOD time series in 2007. After 2008,

there is a small but significant aerosol enhancement (Solomon et al., 2011) in the extratropical lowermost stratosphere due to a series of volcanic eruptions, such as Chaitén, Okmok, Kasatochi, Sarychev, Merapi, Grímsvötn, Nabro, and Puyehue-Cordón Calle (Bourassa et al., 2010; Carn et al., 2016; Haywood et al., 2010) (see Table S1). In contrast to Pinatubo, the aerosol that originated from these later volcanic eruptions rose slowly in the tropics and was mainly transported into the extratropical Northern Hemispheric stratosphere up to an altitude of about 25 km within 1 year. These post-2008 eruptions persistently enhanced the Junge layer in the extratropical lowermost stratosphere (Andersson et al., 2015; Vernier et al., 2011).

3. Effect of Volcanoes on Mean Age and Its Trend

A common metric of the stratospheric circulation is the mean age of stratospheric air (Waugh & Hall, 2002), which is defined as the mean residence time of an air parcel in the stratosphere since its entry across the tropopause. The pattern of mean age trends reveals the pattern of changes of the stratospheric circulation. Our primary focus is on the impact of all volcanic eruptions on the pattern changes of the stratospheric circulation over the past few decades (Haenel et al., 2015) with a particular focus on the Pinatubo eruption, which shows the clearest and most persistent signal in lower stratospheric temperatures (Andersson et al., 2015). Figure 1b shows a clear imprint of the volcanic signal on the deseasonalized mean age during the Pinatubo period. The signal of increasing mean age after the Pinatubo eruption is consistent for simulations driven by both ERA-I and JRA-55 reanalyses.

We quantify the effect of the volcanic aerosol from all volcanic eruptions on mean age and its trends using a multiple linear regression model for the 1989–2012 period. Signals of natural variability such as QBO and El Niño–Southern Oscillation (ENSO) are known to influence the stratospheric circulation and mean age of air (Baldwin et al., 2001; Randel et al., 2009). Therefore, this regression method decomposes the temporal evolution of the monthly zonal mean age in terms of a linear trend, seasonal cycle, QBO, El Niño Southern–Oscillation (ENSO), AOD, and a residual (for details see equation (S1) in section S1 in the supporting information, or Diallo et al., 2012). We disentangle the contribution of the volcanic signal to the mean age calculated from simulations with the Chemical Lagrangian Model of the Stratosphere (CLaMS). CLaMS simulations are driven by ERA-Interim (Dee et al., 2011) and JRA-55 (Kobayashi et al., 2015) meteorological reanalysis fields. In particular, the ERA-Interim-driven CLaMS simulation has been shown to yield mean age largely consistent with spaceborne observations (Ploeger, Riese, et al., 2015). The regression of the mean age is performed both with and without including the term representing the volcanic aerosol signal in the multiple linear regression to isolate the impact of the volcanic aerosol on the mean age and its trend. For representing the mean age variations related to volcanic aerosols, the global AOD time series from merged satellites measurements (Vernier et al., 2011) were used in the multiple linear regression model.

Figures 1c and 1d depicts the residual from the regression (ϵ in equation (S1) in the supporting information) with and without the removal of the AOD signal related to all volcanic eruptions from the mean age. Without removing the AOD signal (blue and black dashed lines), the variability in the residual closely follows the variability in the deseasonalized mean age (Figure 1b). Removing the AOD signal by including the term of volcanic signal in the linear regression significantly reduces the variability in the residual particularly after the Pinatubo eruption (red dashed line in Figures 1c and 1d). Remarkably, including the volcanic aerosol signal in the multiple linear regression substantially reduces the globally averaged mean age linear trend from -0.16 year/decade (without including the AOD signal) to -0.12 year/decade, that is, by about 25% with a p value of 0.2. This demonstrates the robustness of the regression method and corroborates the correlation between the variability in mean age and volcanic aerosols. There is also a reduction in the residual during the volcanically quiescent 1997–2000 period, which is perhaps due to existing correlation between the volcanic eruptions and the strongest ENSO event (1997/2000) (Adams et al., 2003). The reduction of the variability in the residual for the later period after 2008 appears small and is partly canceled due to the global average and the asymmetry of later volcanic eruptions with Northern Hemispheric predominance. Therefore, averaging the effect of the later volcanoes in the Northern Hemisphere shows a significant increase in mean age and reduction of the variability in the residual during the volcanically active period after 2008 (Figure S1).

Figure 2a shows the impact of volcanic aerosol on the stratospheric circulation as estimated from mean age. The mean age changes induced by volcanic aerosol from all volcanic eruptions (1989–2012) are quantified from the term b_3 in equation (S1) normalized by the standard deviation (STD) of the AOD. The dominant signal in the whole AOD time series is associated with the Pinatubo eruption. The figure shows evidence

AGU Geophysical Research Letters



Figure 2. Zonal mean distribution of (a) volcanic aerosol effect on mean age and (b-d) its decadal trends. The amplitude of the mean age variations (term $b_3 \times STD(aod)$) attributed to the volcanic aerosol from all eruptions since Pinatubo eruption are calculated by using AOD signal from the 1989–2012 period. Bold black line indicates the tropopause height. Black contours show the averaged mean age of air in the 1992–1994 period after Pinatubo eruption. Gray dots indicate statistical significance at 95% level estimated from the Student's *t* test for the all volcances. The decadal mean age changes are calculated as a linear trend (term *a*) from the multiple linear regression without including the AOD signal (Figure 2b) and with including the AOD signal (Figure 2c). The difference between the two quantities (without-with) is by definition the volcanic effect on the mean age trends (Figure 2d). This difference is directly comparable to the volcanic effect in Figure 2a, if multiplied with -1, as the Pinatubo eruption occurs at the beginning of the considered period. Figure 2d contours show the fractional change of the trend between Figure 2c–2b.

for an increase of mean age by up to 0.25 year throughout the lower and midstratosphere following the eruption (Figure 2a). The results shown here are based on CLaMS and are analogously reproduced for TRACZILLA simulations in the supporting information (Figure S2). This increase of the mean age did not occur instantaneously but evolved in time as shown in Figure S3. During the first 6 months after the Pinatubo eruption, mean age mainly increased in the extratropical stratosphere above 22 km, while it decreased below that level and also in the deep tropics. The decrease of the mean age indicates increasing upwelling in the deep tropics directly after the eruption, which is consistent with the response in climate models (Garcia et al., 2011). However, after a few months, the increase in mean age spreads into the tropical stratosphere until spring 1993 when the maximum effect of Pinatubo on mean age is observed (Figures 1b and S3). This maximum impact of Pinatubo on mean age arises within the latitude band of approximately 20°S to 30°N and between 20 and 27 km where aircraft and satellite observations of the volcanic plume show a steep meridional gradient of aerosol (McCormick et al., 1995). These results are also consistent with the mean age changes calculated by subtracting the quiescent period (1998 - 2000) from the Pinatubo period (1992 - 1994) (Figure S4a). The difference between the standard deviation of the residual (ϵ in equation (S1) in the supporting information) with and without removal of the AOD signal shows a significant variability in the residual if the volcanic effect is not taken into account in the multiple linear regression (Figure S4b). This changing pattern in the residual is similar to the explicitly calculated volcanic effect for the Pinatubo period (Figure 2a). Therefore, a significant variability remaining in mean age after removing the seasonal cycle, QBO, and ENSO signals can indeed be attributed to stratospheric volcanic aerosol.

The statistical significance of the volcanic aerosol effect on the mean age has been assessed by performing a Student's *t* test. For more detail about the statistical analysis see Diallo et al. (2012) and the supporting information. The gray dots in Figure 2a show the one-sided *p* value of the Student's *t* test for the hypothesis of a null volcanic effect on the mean age. The whole region of a large volcanic effect on the mean age is highly significant. This indicates that the volcanic aerosol effect on mean age in the lower and midstratosphere in Figure 2, which is mainly dominated by the Pinatubo eruption, is statistically robust.

Figures 2b–2d shows the stratospheric circulation changes, measured in terms of mean age trends (linear term *a* in equation (S1) in the supporting information), for the 1989–2012 period, without (Figure 2b) and with (Figure 2c) removing the volcanic aerosol signal in the multiple linear regression. Note that a negative mean age trend corresponds to a speedup of the stratospheric circulation and a positive mean age trend to a slowdown. For the 1989–2012 period, the calculation without including the AOD term in the multiple linear regression (Figure 2b) leads to a more strongly negative trend of the mean age in the Southern and Northern Hemisphere lower stratosphere below 24 km and to a weakened positive trend in the Northern Hemisphere above 24 km than including the AOD term (Figure 2c). The difference between mean age trends with and without removing the AOD signal demonstrates the aging effect of the volcanic aerosol on the mean age (Figure 2d). The trend is more negative if the effect of volcanic eruptions is not removed from the mean age decadal variability. Evidently Pinatubo eruption has a strong impact on mean age trends in the lower and mid stratosphere in the 1989–2012 period, changing the magnitude of the mean age trends by up to 0.2 year/decade (e.g., up to 60%) throughout that region.

An important remaining question concerns the dynamical mechanism involved in the stratospheric circulation changes related to volcanic aerosol injections. The clearest picture can be expected to emerge for the major eruption of Pinatubo on which we concentrate our discussion in the following. An estimate of the response of the tropical upwelling velocity \overline{w}^* to volcanic aerosol injections can be deduced from the latitudinal gradient of mean age of air (e.g., SPARC CCMVal report, chapter 5, Eyring et al., 2010; Strahan et al., 2011). Following the arguments of Neu and Plumb (1999), the extratropics-tropics mean age difference is directly related to the (inverse) upwelling velocity, $\Delta\Gamma \sim 1/\overline{w}^*$, and independent of the mixing strength (see Linz et al., 2016 for a more recent discussion). Figure 3 shows the latitudinal gradient of the net volcanic effect on mean age and its trends (Figure 3a) as well as the \overline{w}^* estimate (Figure 3b). The distinct pattern with a negative effect on the mean age gradient above about 22 km and a positive effect below indicates increasing tropical upwelling above 22 km related to volcanic aerosol (noting $\Delta\Gamma \sim 1/\overline{w}^*$) and decreasing upwelling below. This result agrees well with the \overline{w}^* estimate, which shows an increasing tropical upwelling at the upper levels and decreasing upwelling at the lower levels related to volcanic aerosol (Figure 3b), i.e., a strengthening of the BD circulation (advective) above 22 km and its weakening below. The latitudinal gradient estimated from the net volcanic effect on the mean age trend is positive above 22 km (Figure 3a). This implies a negative volcanic effect on the trend of the tropical ascent rate (noting $d(\Delta\Gamma)/dt \sim -d\overline{w}^*/dt$).

The clear increase of tropical upwelling due to volcanic aerosol above 22 km (Figure 3) implies that the global increase of mean age of air (e.g., Figures 2a and 2d) is likely related to mixing effects. Figure 4 shows the decomposition of globally averaged deseasonalized mean age variability into the residual circulation transit time (RCTT), representing the pure residual circulation effect on mean age, and into aging by mixing, the eddy mixing effect integrated along the residual circulation, following Ploeger, Abalos, et al. (2015) (see supporting information). Directly following the Pinatubo eruption in June 1991, mean age increases significantly throughout the stratosphere above about 20 km (Figure 4a). This increase in mean age can be attributed to both increasing RCTT and increasing aging by mixing, with the RCTT contribution stronger at higher levels above about 22 km (Figures 4b and 4c) and the aging by mixing contribution stronger at all levels. Note that the RCTT represents the integrated residual circulation effect, whereas the \overline{w}^* estimate from the latitudinal age gradient (see above) is a local quantity. Therefore, the increasing RCTT above about 22 km is not in contradiction with the increasing \overline{w}^* estimated from the age latitudinal gradient (Figure 3) but rather indicates that decreasing upwelling at lower levels and pathway changes dominate the RCTT effect. The substantial simultaneous increase in RCTT and aging by mixing indicates a potential link between the integrated mixing effect



Figure 3. Vertical profiles of latitudinal gradient of the net volcanic effect on mean age, its trends, and \overline{w}^* . (a) The vertical profile of the latitudinal gradient of the net volcanic effect on mean age (red) and its trend (black) calculated from Figures 2a and 2d and averaged between the Northern and Southern Hemispheres. (b) Vertical profile of the net volcanic effect on the residual vertical velocities estimated from both standard formula (\overline{w}^*) between 20°S and 20°N over the 1989–2012 period. Green dashed line indicates the altitude 22 km.

and residual circulation changes, as discussed in previous studies (Garny et al., 2014; Pitari et al., 2016; Ray et al., 2014). Changes in the local mixing tendency below about 20 km (not shown) suggest a direct response of eddy mixing to volcanic aerosols at lower levels. Below about 20 km, the decrease in mean age initiated before the Pinatubo eruption extends up to 6 months after eruption. The related decrease in RCTT suggests increasing tropical upwelling directly after the eruption.

Increasing mean age in the tropical lower stratosphere related to volcanic aerosols from reanalysis is consistent with a recent modeling study. Muthers et al. (2016) show increasing mean age in response to volcanic eruptions in the tropical lower stratosphere (20°S-20°N at 63 hPa) and decreasing mean age in the middle and upper stratosphere in Northern Hemisphere high latitudes (60° – 90°N at 16 hPa). Focussing on higher altitudes, other modeling studies also reported mainly decreasing mean age following volcanic eruptions (Garcia et al., 2011; Garfinkel et al., 2017). Similarly, Pitari et al. (2016) found decreasing mean age at higher levels of 30 hPa in the tropics and 10 hPa in the middle latitudes after the Pinatubo eruption. Hence, the response of the shallow BD circulation branch appears qualitatively consistent between ERA-Interim reanalysis and climate models (e.g., SOCOL, Muthers et al., 2016), while the response of the deep branch differs (Garcia et al., 2011; Garfinkel et al., 2017; Muthers et al., 2016). The main difference between the reanalysis and the models' response concerns the depth and strength of the deep versus shallow branch changes. In the reanalysis, the increase of mean age is stronger and reaches higher into the midstratosphere, while in the climate models it is confined to the lower stratosphere. Toohey et al. (2014) found the high-latitude response to be indirectly related to the volcanic heating through changes in wave propagation. Hence, differences in wave propagation (and mixing) between different models are expected to cause differences in the models' circulation response to volcanic eruptions, likely involving a strong role of internal variability, which is also important to the mechanism for changing age (see the ensemble spread in Figure 3 of Toohey et al., 2014).

We note that ERA-Interim and JRA-55 reanalyses do not explicitly assimilate volcanic aerosols. The effect of volcanic aerosols in the reanalysis is entirely related to the assimilation of observed temperatures (Fujiwara et al., 2015), which also constrains the wind response through thermal wind balance. Because of the missing direct

Geophysical Research Letters



Figure 4. Globally deseasonalized time series of the (a) mean age, (b) aging by mixing, and (c and d) residual circulation transit time. Shown are the effects on mean age, aging by mixing, and residual circulation transit time (Figure 4c) without and with (Figure 4d) including assimilation increment related to volcanic eruptions since Pinatubo during the period 1989–2012. The decadal effects are calculated as a residual of the multiple linear regression without including the volcanic AOD signal. Black dashed lines show the height.

aerosol heating, the model underestimates stratospheric temperatures after the eruption and the induced assimilation increment tends to warm the model to fit the observations (see Figure S5). The magnitude of this missing diabatic heating contribution can be estimated from the assimilation increment (see Fueglistaler et al., 2009, for details on calculating the assimilation increment) although the increment includes also other factors. The assimilation increment due to the direct aerosol heating effect is found to enhance the total diabatic heating by about 20% directly following the eruption. It progressively decreases after about 6 months following the eruption (Figure S5a). For the interannual variability, adding the assimilation increment reduces the volcanic signal in the total diabatic heating rate by about 50%, but a significant weakening of tropical upwelling after Pinatubo eruption remains (see Figure S5b). Consequently, including the assimilation increment in the calculation of the RCTT reduces the aging above about 20 km and increases the freshening below right after the eruption (Figure 4d). During the first months after the eruption, the simulation with assimilation

AGU Geophysical Research Letters



Figure 5. Zonal mean distribution of the volcanic aerosol contribution to the decadal mean age trends. Shown are the effects on mean age trends related to the later volcanoes during the period 2002–2011. The decadal mean age changes are calculated as a linear trend from the multiple linear regression (a) without explicitly including the volcanic AOD signal and (b) with including the volcanic signal. (c) The difference between the two quantities (without-with) is by definition the volcanic effect on the mean age trends. Figure 2c contours show the fractional change of the trend between Figure 2a and Figure 2b. Gray dots indicate statistical significance at the 95% level estimated from a Student's *t* test.

increments shows even a strong freshening at upper levels (above about 22 km). Therefore, adding the assimilation increment to the heating rate brings the reanalysis-driven simulation closer to the climate model results. However, even with adding the assimilation increment to correct for the missing direct aerosol heating in the reanalysis, a clear aging after the Pinatubo eruption remains, corroborating our conclusions. Furthermore, reanalysis products may suffer from changes in the assimilated observational data sets (e.g., the introduction of Advanced Microwave Sounding Unit-A data (Dee & Uppala, 2009) in 1998 and of radio occultation data (Poli et al., 2010) at the end of 2006). However, the good agreement between mean age trends from simulations driven by meteorological reanalysis with observations found in previous work provides confidence in the reliability of simulated stratospheric circulation changes (Ploeger, Riese, et al., 2015).

Additionally, Figure 4a shows a significant increase of mean age starting in 2008. This increase in globally averaged mean age can be traced back to a distinct hemispheric trend pattern, with strongly increasing mean age in the Northern Hemisphere and decreasing mean age in the Southern Hemisphere (see Ploeger, Riese, et al., 2015 for details). The pattern of these circulation changes qualitatively agrees with satellite observations and is related to decadal variability (Haenel et al., 2015). Figure 5 shows the effect of the minor volcanic eruptions after 2008 on the stratospheric circulation estimated from the 2002-2011 mean age time series. These minor volcanoes shows a persistent AOD signal in the extratropical lowermost stratosphere (Figure 1a). A multiple linear regression including the observed AOD signal only after 2008 was used to better isolate the effect of extratropical volcanoes and the remaining contributions in the AOD time series set to a background value of 0.0046 nm. Although the post-2008 eruptions are clearly weaker than Pinatubo, the regression method yields a significant aging related to these minor eruptions (see Figure 5c). Remarkably, the volcanic aerosol related effect after 2008 shows a significant increasing mean age in the extratropical lower stratosphere in the Northern Hemisphere and decreasing mean age in the high-latitude Southern Hemisphere during the 2008-2011 period. The strongest impact of the post-2008 eruptions on mean age occurs in the Northern Hemisphere lower stratosphere coinciding with the region of highest extratropical volcanic aerosol loading due to these volcanoes (Andersson et al., 2015; Vernier et al., 2011). Parts of the volcanic plumes of Okmok, Kasatochi, Sarychev, and Nabro were, however, horizontally transported from the extratropical Northern Hemisphere into the tropics (Bourassa et al., 2010; Haywood et al., 2010), where they were slowly lofted within 1 year up to about 25 km. Based on the regression, we hypothesize that the post-2008 minor eruptions significantly change the resulting decadal trend in mean age for the 2002–2011 period (Figure 5c). In large regions of the Northern Hemisphere the slowdown of the stratospheric circulation between 2002 and 2011 can be attributed to the recent minor volcanic eruptions after 2008. The remaining decadal trend in mean age can likely be associated with a southward shift of the stratospheric circulation, as suggested by Stiller et al. (2017). Moreover, potential links between volcanic aerosol effects and stratospheric circulation shift could exist.

Note that decadal mean age changes during 2002–2011 in JRA-55 show a different hemispheric pattern as compared to ERA-Interim and MIPAS observations (not shown), but a similar increase in mean age related to minor volcanic eruptions (Figures 1d and S1d), suggesting that volcanic aerosols are not causing the differences between the two reanalyses.

4. Summary and Conclusions

We have estimated the effect on the stratospheric circulation of increases in stratospheric aerosol loading due to volcanic eruptions via mean age of air and its trends. Mean age of air from model simulations consistent with observations for the 1989–2012 period has been analyzed using a multiple linear regression technique accounting for observed stratospheric aerosol. We find that a substantial contribution in decadal variability of the stratospheric circulation, as represented by variability in mean age of air, is caused by volcanic aerosol injections. Pinatubo clearly stands out impacting the global stratosphere immediately after the eruption. The volcanic aerosol effect on tropical upwelling in the meteorological reanalysis turns out to be qualitatively consistent with climate model results, showing strengthened tropical upwelling at upper levels (above about 22 km) and weakened tropical upwelling below (see Figure 3). The mean age of air globally, whereas climate models show decreasing mean age at upper levels (e.g., Garcia et al., 2011; Garfinkel et al., 2017; Muthers et al., 2016). We find the mean age increase after volcanic eruptions significantly affected by volcanically induced mixing effects. A substantial increase in aging by mixing appears to be related to both an increase in mixing tendency and a slowdown of the residual circulation.

In summary, the main difference between the meteorological reanalyses and climate models' responses concern changes in the depth and strength of the deep versus shallow branch of the stratospheric circulation. In the meteorological reanalyses, the increase of mean age is stronger and reaches higher into the midstratosphere, while in the climate models it is confined to the lower stratosphere (Muthers et al., 2016).

Minor volcanic eruptions after 2008 also show a distinct effect on the stratospheric circulation in the Northern Hemisphere stratosphere, significantly impacting the pattern of decadal mean age variability and its trends during 2002–2011 (Figure 5c). Increases in volcanic aerosol loading after 2008 are not included in current model simulations of recent climate change (Santer et al., 2014; Solomon et al., 2011). We speculate that the inadequate representation of volcanic aerosol forcing, in combination with the lack of a reliable representation of natural variability (e.g., wave propagation, mixing, QBO, and aerosol forcing) could be the reason why current climate models fail in simulating the observed pattern of the stratospheric circulation change over the past few decades in the Northern Hemisphere. The effect on the stratospheric circulation due to volcanoes found here will likely carry over to geoengineering scenarios based on sulfate aerosols injections into the stratosphere (Aquila et al., 2014; Tilmes et al., 2015).

References

- Adams, J. B., Mann, M. E., & Ammann, C. M. (2003). Proxy evidence for an El Niño-like response to volcanic forcing. *Nature*, 426, 274–278. https://doi.org/10.1038/nature02101
- Andersson, S. M., Martinsson, B. G., Vernier, J.-P., Friberg, J., Brenninkmeijer, C. A. M., Hermann, M., ... Zahn, A. (2015). Significant radiative impact of volcanic aerosol in the lowermost stratosphere. *Nature Communications*, *6*, 7692. https://doi.org/10.1038/ncomms8692
- Aquila, V., Garfinkel, C. I., Newman, P. A., Oman, L. D., & Waugh, D. W. (2014). Modifications of the quasi-biennial oscillation by a geoengineering perturbation of the stratospheric aerosol layer. *Geophysical Research Letters*, 41, 1738–1744. https://doi.org/10.1002/2013GL058818
- Baldwin, M. P., Gray, L. J., Dunkerton, T. J., Hamilton, K., Haynes, P. H., Randel, W. J., ... Tkahashi, M. (2001). The quasi-biennial oscillation. *Reviews of Geophysics*, 39, 179–229. https://doi.org/1999RG000073

Bönisch, H. B., Engel, A., Birner, T., Hoor, P., Tarasick, D. W., & Ray, E. A. (2011). On the structural changes in the Brewer-Dobson circulation after 2000. Atmospheric Chemistry and Physics, 11, 3937–3948. https://doi.org/10.5194/acp-11-3937-2011

- Bourassa, A. E., Degenstein, D. A., Elash, B. J., & Llewellyn, E. J. (2010). Evolution of the stratospheric aerosol enhancement following the eruptions of Okmok and Kasatochi: Odin-OSIRIS measurements. *Journal of Geophysical Research*, 115, D00L03. https://doi.org/10.1029/2009JD013274
- Bourassa, A. E., Robock, A., Randel, W. J., Deshler, T., Rieger, L. A., Lloyd, N. D., ... Degenstein, D. A. (2012). Large volcanic aerosol load in the stratosphere linked to Asian monsoon transport. *Science*, 337(6090), 78–81. https://doi.org/10.1126/science.1219371
- Butchart, N. (2014). The Brewer–Dobson circulation. *Reviews of Geophysics*, *52*, 157–184. https://doi.org/10.1002/2013RG000448 Butchart, N., Cionni, I., Eyring, V., Shepherd, T. G., Waugh, D. W., Akiyoshi, H., ... Tian, W. (2010). Chemistry-climate model
 - simulations of twenty-first century stratospheric climate and circulation changes. *Journal of Climate*, 23(20), 5349–5374. https://doi.org/10.1175/2010JCLI3404.1

Carn, S. A., Clarisse, L., & Prata, A. J. (2016). Multi-decadal satellite measurements of global volcanic degassing. *Journal of Volcanology and Geothermal Research*, 311, 99–134. https://doi.org/10.1016/j.jvolgeores.2016.01.002

Acknowledgments

We particularly thank the ECMWF and JMA for providing reanalyses data set. This study was supported by the Helmholtz Association under grant VH-NG-1128 (Helmholtz-Hochschul-Nachwuchsforschergruppe), a research stay at the Institute of Energy and Climate Research, Stratosphere (IEK-7), Forschungszentrum in Jülich. The French Labex "Étude des géofluides et des VOLatils-Terre, Atmosphère et Interfaces-Ressources et Environnement" (VOLTAIRE) (ANR-10-LABX-100-01) managed by the University of Orleans has also supported this study at each earlier stage. The authors want also to thank Jean-Paul Vernier (NASA US) for providing the merged SAGE II, GOMOS, and CALIPSO aerosol optical depth and Landon Rieger and Adam Bourassa (USASK CA) for the SAGE II and OSIRIS AOD. The AOD data are not publicly available but can be obtained by contacting Jean-Paul Vernier (jeanpaul.vernier@nasa.gov). The QBO and MEI are available from NOAA website (http://www.cpc.ncep.noaa. gov/data/indices/gbo.u30.index) and (https://www.esrl.noaa.gov/psd/ enso/mei/). Finally, our thanks go to the anonymous reviewers. The TRACZILLA mean age data can be requested from the author (m.diallo@fz-iuelich.de) and CLaMS mean age data set from the coauthor F. Ploeger (f.ploeger@fz-juelich.de).

Dee, D. P., & Uppala, S. (2009). Variational bias correction of satellite radiance data in the era-interim reanalysis. Quarterly Journal of the Royal Meteorological Society, 135, 1830–1841. https://doi.org/10.1002/qj.493

Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., ... Vitart, F. (2011). The Era-Interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137, 553–597. https://doi.org/10.1002/qj.828

Diallo, M., Legras, B., & Chédin, A. (2012). Age of stratospheric air in the Era-Interim. Atmospheric Chemistry and Physics, 12, 12,133–12,154. https://doi.org/10.5194/acp-12-12133-2012

Diallo, M., Legras, B., Ray, E., Engel, A., & Añel, J. A. (2017). Global distribution of CO₂ in the upper-troposphere and stratosphere. Atmospheric Chemistry and Physics, 17(6), 3861–3878. https://doi.org/10.5194/acp-17-3861-2017

Engel, A., Möbius, T., Bönisch, H., Schmidt, U., Heinz, R., Levin, I., ... Boering, K. (2009). Age of stratospheric air unchanged within uncertainties over the past 30 years. *Nature Geoscience*, *2*, 28–31. https://doi.org/10.1038/ngeo388

Eyring, V., Shepherd, T. G., & Waugh, D. W. (2010). Sparc CCMVal report on the evaluation of chemistry-climate models. In V. Eyring, T. Shepherd & D. Waugh (Eds.) (SPARC Report No. 5, WCRP-30/2010, WMO/TD - No. 40, Vol. 5, 426 pp.). SPARC Office. Retrieved from http://www.sparc-climate.org/publications/sparc-reports/

Fromm, M., Kablick III, G., Nedoluha, G., Carboni, E., Grainger, R., Campbell, J., & Lewis, J. (2014). Correcting the record of volcanic stratospheric aerosol impact: Nabro and Sarychev Peak. *Journal of Geophysical Research: Atmospheres*, 119, 10,343–10,364. https://doi.org/10.1002/2014JD021507

Fueglistaler, S., Dessler, A. E., Dunkerton, T. J., Folkins, I., Fu, Q., & Mote, P. W. (2009). Tropical tropopause layer. *Reviews of Geophysics*, 47, RG1004. https://doi.org/10.1029/2008RG000267

Fujiwara, M., Hibino, T., Mehta, S. K., Gray, L., Mitchell, D., & Anstey, J. (2015). Global temperature response to the major volcanic eruptions in multiple reanalysis data sets. Atmospheric Chemistry and Physics, 15(23), 13,507–13,518. https://doi.org/10.5194/acp-15-13507-2015

Garcia, R. R., Randel, W. J., & Kinnison, D. E. (2011). On the determination of age of air trends from atmospheric trace species. *Journal of Atmospheric Science*, 68, 139–154. https://doi.org/10.1175/2010JAS3527.1

Garfinkel, C. I., Aquila, V., Waugh, D. W., & Oman, L. D. (2017). Time-varying changes in the simulated structure of the Brewer–Dobson circulation. *Atmospheric Chemistry and Physics*, 17(2), 1313–1327. https://doi.org/10.5194/acp-17-1313-2017

Garny, H., Birner, T., Bönisch, H., & Bunzel, F. (2014). The effects of mixing on age of air. Journal of Geophysical Research: Atmospheres, 119, 7015–7034. https://doi.org/10.1002/2013JD021417

Garny, H., Dameris, M., Randel, W., Bodeker, G. E., & Deckert, R. (2011). Dynamically forced increase of tropical upwelling in the lower stratosphere. *Journal of Atmospheric Science*, 68, 1214–1233. https://doi.org/10.1175/2011JAS3701.1

Haenel, F. J., Stiller, G. P., von Clarmann, T., Funke, B., Eckert, E., Glatthor, N., ... Reddmann, T. (2015). Reassessment of MIPAS age of air trends and variability. Atmospheric Chemistry and Physics, 15(22), 13,161–13,176. https://doi.org/10.5194/acp-15-13161-2015

Hall, T., & Plumb, R. A. (1994). Age as a diagnostic of stratospheric transport. *Journal of Geophysical Research*, *99*, 1059–1070. Hansen, J., Sato, M., Ruedy, R., Lacis, A., Asamoah, K., Borenstein, S., ... Wilson, H. (1996). A Pinatubo climate modeling investigation.

In G. Fiocco, D. Fua, & G. Visconti (Eds.), *The Mount Pinatubo eruption: Effects on the atmosphere and climate* (Vol. I 42, pp. 233–272), Nato ASI Series. Heidelberg, Germany: Springer.

Haywood, J. M., Jones, J., Clarisse, L., Bourassa, A., Barnes, J., Telford, P., ... Braesicke, P. (2010). Observations of the eruption of the Sarychev volcano and simulations using the HadGEM2 climate model. *Journal of Geophysical Research*, *115*, D21212. https://doi.org/10.1029/2010JD014447

Huber, M., & Knutti, R. (2014). Natural variability, radiative forcing and climate response in the recent hiatus reconciled. *Nature Geoscience*, 7(7692), 651–656. https://doi.org/10.1038/ngeo2228

Junge, C. E., & Manson, J. E. (1961). Stratospheric aerosol studies. Journal of Geophysical Research, 66(7), 2163–2182. https://doi.org/10.1029/JZ066i007p02163

Kobayashi, S., Ota, Y., Harada, Y., Ebita, A., Boriya, M., Onoda, H., ... Takahashi, K. (2015). The JRA-55 reanalysis: General specifications and basic characteristics. *Journal of the Meteorological Society of Japan*, 93(1), 5–48. https://doi.org/10.2151/jmsj.2015-001

Legras, B., Pisso, I., Berthet, G., & Lefevre, F. (2005). Variability of the Lagrangian turbulent diffusion in the lower stratosphere. *Atmospheric Chemistry and Physics*, *5*(6), 1605–1622. https://doi.org/10.5194/acp-5-1605-2005

Linz, M., Plumb, R. A., Gerber, E. P., & Sheshadri, A. (2016). The relationship between age of air and the diabatic circulation of the stratosphere. Journal of the Atmospheric Sciences, 73(11), 4507–4518. https://doi.org/10.1175/JAS-D-16-0125.1

Mahieu, E., Chipperfield, M. P., Notholt, J., Reddmann, T., Anderson, J., Bernath, P. F., ... Walker, K. A. (2014). Recent Northern Hemisphere stratospheric HCI increase due to atmospheric circulation changes. *Nature*, 515(5542), 104–107. https://doi.org/10.1038/nature13857 McCormick, M. P., Thomason, L. W., & Trepte, C. R. (1995). Proxy evidence for an El Niño-like response to volcanic forcing. *Nature*, 373,

399–404. https://doi.org/10.1038/373399a0
Muthers, S., Kuchar, A., Stenke, A., Schmitt, J., Anet, J. G., Raible, C. C., & Stocker, T. F. (2016). Stratospheric age of air variations between 1600 and 2100. *Geophysical Research Letters*, 43, 5409–5418. https://doi.org/10.1002/2016GL068734

Neu, J. L., & Plumb, R. A. (1999). Age of air in a "leaky pipe" model of stratospheric transport. Journal of Geophysical Research, 104(D19), 19,243–19,255. https://doi.org/10.1029/1999JD900251

Pitari, G., Cionni, I., Di Genova, G., Visioni, D., Gandolfi, I., & Mancini, E. (2016). Impact of stratospheric volcanic aerosols on age-of-air and transport of long-lived species. Atmosphere, 7(11), 149.

Ploeger, F., Abalos, M., Birner, T., Konopka, P., Legras, B., Müller, R., & Riese, M. (2015). Quantifying the effects of mixing and residual circulation on trends of stratospheric mean age of air. *Geophysical Research Letters*, 42, 2047–2054. https://doi.org/10.1002/2014GL062927

Ploeger, F., Riese, M., Haenel, F., Konopka, P., Müller, R., & Stiller, G. (2015). Variability of stratospheric mean age of air and of the local effects of residual circulation and eddy mixing. *Journal of Geophysical Research: Atmospheres*, 120, 716–733. https://doi.org/10.1002/2014JD022468

Poli, P., Healy, S. B., & Dee, D. P. (2010). Assimilation of global positioning system radio occultation data in the ECMWF ERA-Interim reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 136, 1972–1990. https://doi.org/10.1002/qj.722

Pommrich, R., Müller, R., Grooβ, J.-U., Konopka, P., Ploeger, F., Vogel, M. T. B., ... Riese, M. (2014). Tropical troposphere to stratosphere transport of carbon monoxide and long-lived trace species in the Chemical Lagrangian Model of the Stratosphere (CLaMS). *Geoscientific Model Development*, *7*, 2895–2916. https://doi.org/10.5194/gmd-7-2895-2014

Randel, W. J., Garcia, R. R., Calvo, N., & Marsh, D. (2009). ENSO influence on zonal mean temperature and ozone in the tropical lower stratosphere. *Geophysical Research Letters*, 39, L15822. https://doi.org/10.1029/2009GL039343

Ray, E. A., Moore, F. L., Rosenlof, K. H., Davis, S. M., Sweeney, C., & Bnisch, H. (2014). Improving stratospheric transport trend analysis based on SF₆ and CO₂ measurements. *Journal of Geophysical Research*, *119*, 14,110–14,128. https://doi.org/10.1002/2014JD021802

- Riese, M., Ploeger, F., Rap, A., Vogel, B., Konopka, P., Dameris, M., & Forster, P. (2012). Impact of uncertainties in atmospheric mixing on simulated UTLS composition and related radiative effects. *Journal of Geophysical Research*, 117, D16305. https://doi.org/10.1029/2012JD017751
- Santer, B. D., Bonfils, C., Painter, J. F., Zelinka, M. D., Mears, C., Solomon, S., ... Wentz, F. J. (2014). Volcanic contribution to decadal changes in tropospheric temperature. *Nature Geoscience*, 7(7692), 185–189. https://doi.org/10.1038/ngeo2098
- Sato, M., Hansen, J. E., McCormick, M. P., & Pollack, J. B. (1993). Stratospheric aerosol optical depth. *Journal of Geophysical Research*, 98(D12), 22,987–22,994.
- Schmidt, G. A., Shindell, D. T., & Tsigaridis, K. (2014). Reconciling warming trends. Nature Geoscience, 7(7692), 158–160. https://doi.org/10.1038/ngeo2105
- Solomon, S., Daniel, J. S., Neely, R. R., Vernier, J. P., Dutton, E., & Thomason, L. W. (2011). The persistently variable "background" stratospheric aerosol layer and global climate change. *Science*, 333(6044), 866–870. https://doi.org/10.1126/science.1206027
- Solomon, S., Rosenlof, K. H., Portmann, R. W., Daniel, S. M., Davis, J. S., Sanford, T., & Plattner, G.-K. (2010). Contributions of stratospheric water vapor to decadal changes in the rate of global warming. *Science*, *327*(5970), 1219–1223. https://doi.org/10.1126/science.1182488
- Stiller, G. P., Fierli, F., Ploeger, F., Cagnazzo, C., Funke, B., Haenel, F. J., ... von Clarmann, T. (2017). Shift of subtropical transport barriers explains observed hemispheric asymmetry of decadal trends of age of air. *Atmospheric Chemistry and Physics Discussions*, 2017, 1–16. https://doi.org/10.5194/acp-2016-1162
- Stiller, G. P., von Clarmann, T., Haenel, F., Funke, B., Glatthor, N., Grabowski, U., ... López-Puertas, M. (2012). Observed temporal evolution of global mean age of stratospheric air for the 2002 to 2010 period. *Atmospheric Chemistry and Physics*, *12*(7), 3311–3331. https://doi.org/10.5194/acp-12-3311-2012
- Strahan, S. E., Douglass, A. R., Stolarski, R. S., Akiyoshi, H., Bekki, S., Braesicke, P., ... Yamashita, Y. (2011). Using transport diagnostics to understand chemistry climate model ozone simulations. *Journal of Geophysical Research*, *116*, D17302. https://doi.org/10.1029/2010JD015360
- Tilmes, S., Mills, M. J., Niemeier, U., Schmidt, H., Robock, A., Kravitz, B., ... English, J. M. (2015). A new Geoengineering Model Intercomparison Project (GeoMIP) experiment designed for climate and chemistry models. *Geoscientific Model Development*, 8(1), 43–49. https://doi.org/10.5194/gmd-8-43-2015
- Tilmes, S., Müller, R., & Salawitch, R. (2011). The sensitivity of polar ozone depletion to proposed geoengineering schemes. *Science*, 320(5880), 1201–1204. https://doi.org/10.1126/science.1153966
- Toohey, M., Krüger, K., Bittner, M., Timmreck, C., & Schmidt, H. (2014). The impact of volcanic aerosol on the northern hemisphere stratospheric polar vortex: Mechanisms and sensitivity to forcing structure. *Atmospheric Chemistry and Physics*, 14(23), 13,063–13,079. https://doi.org/10.5194/acp-14-13063-2014
- Trepte, C. R., & Hitchman, M. H. (1992). Tropical stratospheric circulation deduced from satellite aerosol data. *Nature*, 355(0028-0836), 626-628. https://doi.org/10.1038/355626a0
- Vernier, J. P., Thomason, L. W., Pommereau, J. P., Bourassa, A., Pelon, J., Garnier, A., ... Vargas, F. (2011). Major influence of tropical volcanic eruptions on the stratospheric aerosol layer during the last decade. *Geophysical Research Letters*, 38, L12807. https://doi.org/10.1029/2011GL047563
- Waugh, D., & Hall, T. (2002). Age of stratospheric air: Theory, observations, and models. *Reviews of Geophysics*, 40(4), 1010. https://doi.org/10.1029/2000RG000101